Author’s response to reviews

Title: Risk factors for eczema in infants born in Cuba: a population-based cross-sectional study

Authors:

Ramon Suarez-Medina (ramonsm@inhem.sld.cu)
Silvia Venero-Fernandez (silvias@infomed.sld.cu)
Esperanza Mora-Faife (mora@infomed.sld.cu)
Ileana Valle-Infante (idelvalle@infomed.sld.cu)
Gladys Garcia-Garcia (gladysgarcia@informed.sld.cu)
Liem Gomez-Marreo (liemgomez@infomed.sld.cu)
Dania Fabre-Ortiz (fabre@infomed.sld.cu)
Hermes Fundora-Hernandez (hermes.fundora@infomed.sld.cu)
Andrea Venn (andrea.venn@nottingham.ac.uk)
John Britton (j.britton@virgin.net)
Andrew Fogarty (andrew.fogarty@nottingham.ac.uk)

Version: 2 Date: 27 November 2013

Author’s response to reviews:

1. We have added the details of the consent specifically as requested.

2. The University of Nottingham code for payment of this manuscript is IN-1138271. I am unable to enter this into your system but this code will be required if the manuscript is accepted.